Author’s response to reviews

Title: Factors affecting the early failure of implants placed in a dental practice with a specialization in implantology – a retrospective study

Authors:

Johannes Krisam (johannes.krisam@imbi.uni-heidelberg.de)
Larissa Ott (larissa.ott@med.uni-heidelberg.de)
Stephanie Schmitz (dr.stephanie.schmitz@web.de)
Anna-Luisa Klotz (anna-luisa.klotz@med.uni-heidelberg.de)
Aida Seyidaliyeva (aida.seyidaliyeva@med.uni-heidelberg.de)
Peter Rammelsberg (peter.rammelsberg@med.uni-heidelberg.de)
Andreas Zenthöfer (andreas.zenthoefer@med.uni-heidelberg.de)

Version: 3 Date: 21 Aug 2019

Author’s response to reviews:

1. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Response: Done and a list of abbreviations was added to the declarations section.

2. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response: This was ensured and a statement was added to the respective section.

3. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: Thank you. The manuscript was uploaded in a clean version. No figures or additional files.